

Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 11732

Students' Behavioural Responses to a Fallback Option: Evidence from Introducing Interim Degrees in German Schools

Natalie Obergruber Larissa Zierow

AUGUST 2018



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 11732

Students' Behavioural Responses to a Fallback Option: Evidence from Introducing Interim Degrees in German Schools

Natalie Obergruber ifo Institute and IZA

Larissa Zierow ifo Institute and University of Munich

AUGUST 2018

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

	IZA – Institute of Labor Economics	
Schaumburg-Lippe-Straße 5–9 53113 Bonn, Germany	Phone: +49-228-3894-0 Email: publications@iza.org	www.iza.org

ABSTRACT

Students' Behavioural Responses to a Fallback Option: Evidence from Introducing Interim Degrees in German Schools*

Without a school degree, students can have difficulty in the labor market. To improve the lives of upper-secondary school dropouts, German states instituted a school reform that awarded an interim degree to high-track students upon completion of Grade 9. Using retrospective spell data on school and labor market careers from the National Educational Panel Study (NEPS), our difference-in-differences approach exploits the staggered implementation of this reform between 1965 and 1982. As intended, the reform reduced switching between school tracks. Surprisingly, it also increased successful hightrack completion, university entrance rates, and later income, arguably by reducing the perceived risk of trying longer in the high-track school.

JEL Classification:	120, 124, 128
Keywords:	school dropout, school degree, school tracking

Corresponding author:

Natalie Obergruber ifo Center for the Economics of Education Poschingerstr. 5 81679 Munich Germany E-mail: obergruber@ifo.de

^{*} We would like to thank Ludger Woessmann, Uwe Sunde and Derya Uysal for helpful comments on previous versions of this draft. Financial support by the German Research Foundation (DFG) through Priority Program (SPP) 1646 "Education as a Lifelong Process" and by NORFACE through project "The impact of childhood circumstances on individual outcomes over the life-course" is gratefully acknowledged. This paper uses data from the National Educational Panel Study (NEPS): Starting Cohort Adults, doi:10.5157/NEPS:SC6:8.0.0. From 2008 to 2013, NEPS data was collected as part of the Framework Program for the Promotion of Empirical Educational Research funded by the German Federal Ministry of Education and Research (BMBF). As of 2014, NEPS is carried out by the Leibniz Institute for Educational Trajectories (LIfBi) at the University of Bamberg in cooperation with a nationwide network.

1 Introduction

In 1967, sociologists Ralf Dahrendorf and Hansgert Peisert published an influential report on dropouts from German high-track schools (*Gymnasiums*), which, with its finding that about 20% of high-track students drop out, alarmed politicians. In response, the Standing Conference of the Ministers of Education and Cultural Affairs of the Länder agreed to reform its high-track degree policy. The goal was to support dropouts from the high-track, who, although probably of high ability, struggled to enter the labor market due to lack of a school degree. Some states enacted the reform right away; others did so much later. The East German state Thuringia was the last to implement the reform, and did so only after a high-track school dropout went on a shooting rampage in the city of Erfurt in 2002. This tragedy led to media coverage of the state's failure to institute the reform, eventually pressuring politicians to act.

In this paper, we analyze this controversial reform of the German school system and find that it led to unintended behavioral responses by high-track students. Germany's school system features tracking and grade repetition. The school reform under study awarded high-track dropouts, that is, those who failed to complete 13 years of schooling, with a low-track school degree if they finished at least the ninth grade.¹ Before the reform, students who dropped out of the high track did not receive any degree and thus had to qualify for the labor market otherwise, for example, by switching to a low-track school and repeating a grade there. After the reform, students received a basic school degree if they finished ninth grade in the high track, and received a middle school degree if they completed Grade 10 of the high track.

Theoretically, such a reform could lead to various outcomes via the following four mechanisms. The first possible outcome is that the number of students who leave the school system without any school degree should decrease. This was the main goal of the reform. If high-track dropouts switched schools to obtain any other low-track school degree, the reform should result in reduced downgrading. Moreover, being awarded at least some sort of degree, regardless of time spent in school, should improve the students' labor market prospects in terms of em-

¹The high track in Germany is the most time-consuming secondary school track, which grants students a degree at the end of Grade 13. In contrast, in the low tracks (*Hauptschule*, basic school, and *Realschule*, middle school) students obtain a low-track school degree after having finished either Grade 9 or 10. In recent years, the number of years of schooling required in the high track was reduced to 12. This change, however, took place much later than the reform under study.

ployment and wages (see Layard & Psacharopoulos 1974, Hungerford & Solon 1987, Jaeger & Page 1996).

The second and third possible outcomes involve incentives for students at risk of dropping out. On the one hand, the reform incentivizes students to leave the high-track school, and enter the labor market, after having completed either Grade 9 or 10. This would result in a lower number of students with a high-track degree, a higher number of students with a low-track degree, and younger labor market entry age.

On the other hand, the reform reduced the risk of staying in the high track. Parents who might have favored downgrading their low-performing child from the high track to the low track before the reform might now be more likely to keep the child in the high track. Given that the child will be awarded a low-track degree after successful completion of ninth grade, interpretable as *interim degree*, the reform reduced the risk of staying in the high track (on the relevance of risk aversion for schooling decisions, see Lavecchia, Liu & Oreopoulos 2016, Woelfel & Heineck 2012). Thus, students could be motivated to try hard to successfully finish the high track. This behavioral response to the reform would lead to an increased completion rate of the high track and, eventually, to a higher university enrollment rate.

The fourth possible outcome of the reform is that after it has been implemented and students and their parents are aware of it, selection into the high track could increase. The possibility of receiving a (low-track) school degree in the high track, even in case of failure after the ninth grade, reduces the risk of high-track enrollment. Therefore, parents who might have sent their low-performing child to a low-track school before the reform might now take these new opportunities into account when deciding on their child's secondary school, which could lead to an increase in the high-track school enrollment rate.

Our study estimates the causal effects of the German high-track reform and thus discovers which of the above presented outcomes is the most prevalent. For identification we exploit the different timing of reform implementation between 1965 and 1999 across 11 West German states² employing a difference-in-differences (DiD) framework. We compare high-track school students in a reform-implementing state before and after the reform and compare the changes in

²We do not include the East German states in our analysis as their education system underwent substantial change after Reunification.

student outcomes to the changes in high-track school student outcomes in states that did not reform at the same point in time. The inclusion of state fixed effects allows furthermore to abstain from any state-specific features that could affect student outcomes and the inclusion of time fixed effects controls for potential nation-wide shocks. Note that, in the baseline specification, we restrict the sample to students who were already enrolled in high-track schools at the time of the reform in order to rule out reform-induced selection into high-track schools.

For estimation, we use rich and detailed survey data from the National Educational Panel Study (NEPS) adult cohort from 2010 (for more detail on the data, see Blossfeld, Rossbach & von Maurice 2011). The NEPS data have two specific features that enable us to tackle our research question in a methodologically clear-cut manner. First, the data include information on the nature of dropping-out: that is, we can differentiate between whether a student left school for good, whether the student switched to another school (i.e., another school track), and at which point in time the student effectively ended her schooling career.³ Second, the NEPS data contain information on the state in which the student attended school as well as his or her state of current residence. This information is very important for an analysis of state reforms in a context where migration across states during an individual's lifespan is possible.

Our results show that the reform was successful, at least in the sense that students are less likely to switch between schools and tracks. This reform effect is thus in line with the first theoretical consideration. The results, however, do not support the second theoretical consideration. That is, we do not find that more students leave school with a low-track school degree to enter the labor market earlier. Instead, we find that students, rather surprisingly, stay in the school system longer and finish school with a higher school degree due to the reform. This supports the third theoretical consideration. The reform seems to reduce the perceived risk of trying longer in the high-track and thus leads to an increase in the high-track completion rate and in university enrollment rates. In line with this finding, we find significant reform effects on net household income in 2010. We hardly find evidence for the forth theoretical consideration as the likelihood of enrolling into the high track does not significantly increase after

³This is in contrast to other studies that rely on school data reporting how many students left a school without degree but with no information as to the subsequent educational path of the reported dropout. Usually, dropout is not well defined as students are not tracked over time.

the reform. When it comes to equal opportunity effects of the reform, we find that students from a low socioeconomic background benefit more from the reform than students from a high socioeconomic background.

Our study relates to four strands of the literature. First, similar to a growing literature in the economics of education, we employ the DiD approach to evaluate how an education policy reform affects student outcomes. Leading examples of DiD applications that exploit differences in the timing of educational reform implementation across regions within a country include the following: Acemoglu & Angrist (2001) evaluating compulsory schooling laws in the United States, Meghir & Palme (2005) studying a comprehensive school reform in Sweden, Pekkarinen, Uusitalo & Kerr (2009) looking at a tracking reform in Finland, Havnes & Mogstad (2011) estimating the effects of a child-care expansion reform in Norway, and Eyles, Hupkau & Machin (2016) studying a school autonomy reform in England. The DiD approach has also been used to study reforms of the German school system. Pischke (2007) and Pischke & von Wachter (2008), for example, studied German compulsory schooling reforms.⁴ Analogously to this literature, we, too, make use of variation in the time of implementing the high track reform across states and time.

Our focus on a reform that changed the opportunities of (potential) drop-out students in the high track contributes to the literature on tracking and on dropouts. The extant literature on tracking finds tracking systems to increase educational inequality, for example, Piopiunik (2014), Hanushek & Woessmann (2006), yet there is no evidence to date on how the possibility of being granted an interim degree from a high-track school influences the number of years individuals stay in school. It might be that especially risk-averse students try to finish the high track when they are offered an interim degree, in which case the reform under study may be working as a remediating tool in a school system that uses tracking, similar to the benefits of up- and downgrading in a tracked system found by Dustmann, Puhani & Schoenberg (2017). The literature on school dropouts frequently focuses on dropouts from the low track (i.e., lower

⁴German compulsory schooling reforms have also been studied by Siedler (2010), Piopiunik (2014), and Kamhoefer & Schmitz (2016). Another German school reform has been studied by Riphahn (2012), who investigates the abolition of upper-track school fees in 1950. A recent reform shortening the number of required years in high track schools has been studied by, among others, Thiel, Thomsen & Buettner (2014), Dahmann & Anger (2015), Andrietti (2015), and Huebener & Marcus (2017).

secondary school) who often end up without any school degree and leave the school system for good (e.g., Oreopoulos (2007); Anderson (2014); Dearden, Emmerson & Meghir (2009)). Our study, in contrast, focuses on the fate of high-ability high-track dropouts, which is more comparable to school career interruption studied by DesJardins, Ahlburg & McCall (2006). The latter find that college students who have dropped out once are more likely to experience subsequent enrollment disruptions, which are found to be detrimental to the students' probability of graduating. We contribute to the dropout literature because we can reconstruct the whole school and labor market careers of individuals surveyed in our data and link these developments to reforms on the state level. We are thus able to distinguish between real dropouts and individuals who are simply interrupting their schooling career.

The fourth strand of literature to which we contribute is the emerging literature on behavioral aspects of the economics of education. One study closely related to ours is Heckman, Humphries, LaFontaine & Rodriguez (2012), who look at the GED Testing Program in the United States. This program allows dropouts to attain a degree via the GED, which is a high school diploma equivalent. Although, no doubt, this program was begun with good intentions, the authors find that it actually created incentives to drop out from high school. These findings show the importance of considering the riskiness of decisions in the educational system when investigating student behavior. In the same vein, some other studies show that students' (and their parents') risk aversion and time preferences are important factors in the education production function (e.g., Golsteyn, Groenqvist & Lindahl (2014); Woelfel & Heineck (2012)). Consequently, behavioral incentives in the education system that address these factors can play an important role in educational careers and labor market outcomes. For example, Levitt, List, Neckermann & Sadoff (2016) show that student effort is increased by immediate rewards for performance (for an extensive overview of behavioral economics of education, see Lavecchia et al. 2016). In view of this literature, the German school reform under study that introduced an interim degree in the high track can be interpreted as a low-cost intervention that encouraged trying longer in the high track.

This study proceeds as follows. In Section 2 we provide information on the institutional background of the German school system and, in particular, on the reform under study. Section

3 presents the main data sets and provides descriptive statistics. Section 4 explains the empirical framework. Section 5 presents the results and Section 6 investigates the reform effects on the equality of opportunity. Section 7 discusses a series of robustness checks and potential mechanisms. Section 8 concludes.

2 Institutional Background

The institutional set up of Germany's education system permits a detailed analysis of the dropout reform's effects on high-track school students' behavior, including school switching, track down- and upgrading, and grade repetition. In the following, we provide a brief overview of the German school system as it existed for the West German birth cohorts under study and describe the reform and its timing in detail.

The German School System Germany's school system is decentralized, meaning that each of the country's 16 states is responsible for the education of its youth. Although there are some differences across states, the general structure is fairly uniform and illustrated in Figure 1. The school system tracks children at age 10, after four years of primary school, into mainly three secondary school tracks.⁵ The different tracks are characterized by the academic requirements and the final degree awarded. In the basic school (Hauptschule), a student must pass five grades before compulsory schooling ends at ninth grade. The basic school education terminates in a basic school degree that, traditionally, enables students to go on to vocational training (Helbig & Nikolai 2015). In the middle school (Realschule), a student must pass six grades and finishes with a middle school degree. The middle school degree opens up more vocational training opportunities. In the high track (Gymnasium), students must pass nine grades and finish with an academic school degree (Abitur), which is the university entrance qualification.⁶ Tracking is based on ability. Teachers in primary school recommend the highest school track they think is suitable for the child. In some states, this recommendation acts as a top limit to the schooling

⁵In two states, Berlin and Brandenburg, tracking takes place at age 12 after six years of primary school.

⁶Helbig & Nikolai (2015) provide insight into various dimensions of Germany's school system, including at which stage each type degree can be obtained.

available to the child. Parents have the responsibility of choosing the child's secondary school track from the (limited) set of available school tracks. Upgrading from lower tracks into higher ones is generally possible when the attended lower track is completed successfully. Students may also downgrade to a lower track at any time. After secondary school, individuals may either go into vocational training, which consists of classes at a vocational school as well as practical training from an employer, or enter tertiary education.⁷

In evaluating the reform it is important to remember that high-ability children are tracked into the high track. Nonetheless, it is possible to dropout before obtaining a school degree. The high track takes the longest to accomplish and is the most demanding academically.

The Reform Under Study In the 1950s and 1960s there was no legal regulation of the school degree high-track students could receive when dropping out from the high track.⁸ Figure 2 shows that close to 10% of all students who attend the eighth grade in high-track schools drop out in all birth cohorts from 1950 onward; in Baden-Württemberg, during the same decades, more than 20% of students dropped out from the high track without a degree (Peisert & Dahrendorf 1967). The first regulations in regard to degrees for dropouts solely dealt with those students who did not achieve the high-track school degree in 13th grade. It was not until 1965 that regulations on degrees for dropouts from lower grades began to be discussed. In 1965, the *Kultus-ministerkonferenz* (the Standing Conference of the Ministers of Education and Cultural Affairs of the Länder) announced a reform making school-leaving diplomas from the high track comparable to school degrees from lower tracks.⁹ The goal of the reform was to make vocational training available to dropouts. Although the reform was announced at the federal level in 1965,

⁷See Biewen & Tapalaga (2017) and Buchholz & Schier (2015) for a good overview and description of school careers in the German school system using the NEPS adult cohort data.

⁸There is some evidence for Baden-Wüerttemberg that the custom was to count certificates for having passed 10th grade as a middle school degree, at least for some occupations, for example, in the public sector. Other states apparently did not engage in the same practice.

⁹The original name of the reform was "Gleichwertigkeit von Abgangszeugnissen der Realschulen und Gymnasien mit Abschlusszeugnissen der Hauptschulen" ("Equality of School Leaving Diploma of Middle and High-Track Schools with Basic School Degrees"). Thus, students in the middle track schools were also subject to the reform. Additionally, changes in the school careers of high- and middle-track students might have had spillover effects on students in the low-track schools. We examine this issue in more detail in Section 7.

the German states enacted it at different points in time; see Table 1 (Helbig & Nikolai 2015). The first state to reform was Baden-Württemberg, Bavaria reformed only in 1982, and the states of East Germany reformed after Reunification. Thuringia was the last to implement the reform and did so only after a high-track school dropout went on a shooting rampage in the city of Erfurt in 2002. This tragedy led to media coverage of the state's failure to institute the reform, eventually pressuring politicians to act. In some states (e.g., Berlin), the reform granted low-track school degrees retroactively.

The reform aimed at supporting those students who, although probably of high ability, struggled to enter the labor market due to lack of a school degree. Previously, students generally had to obtain a low-track school degree via some other means before entering the labor market. Our analysis focuses on implementation of the reform in West Germany between 1965 and 1982. We exclude the former East German states as the school systems in this section of the country were not completely comparable during the period of study. Students from Bremen, where the reform took place in 1999, are observed only in the untreated state.

As we are interested in evaluating whether the reform improved the situation for dropouts, we would like to know what students' outside options were when dropping out of school without a degree. As dropouts are rarely tracked over time, there is little knowledge about pre-reform outside options. There is some narrative evidence from Peisert & Dahrendorf (1967) in regard to the state of Baden-Württemberg. Peisert & Dahrendorf (1967) interview students from high-track schools and track dropouts for some years after they left the high track. They find that the majority of dropouts are still going to school during the years after dropout in an attempt to obtain either a middle- or even a high-track school degree. The second biggest group of dropouts are employed in public service or trade occupations that count a 10th grade certificate as a middle school degree. Peisert & Dahrendorf (1967) mention the following reasons for dropout in the early 1960s before obtaining a high-track school degree: personal reasons, problems in school subjects, family reasons, the school, teachers, the peer group in school, and commuting to school. Personal reasons include that students do not want to go to school anymore or that

they have decided on an occupation for which an academic degree is not necessary. Problems in school subjects was the most frequently stated reason for dropout.

3 Data

We combine two unique data sources to estimate the reform effects. First, we make use of rich and detailed survey data from the adult cohort of the National Educational Panel Study (NEPS) from 2010.¹⁰ We extract attendance at the high track without achieving a school degree and our education-related outcomes from spell data on individual schooling histories. Second, we use the reform years reported in Table 1 as the starting point for the treatment. We create a treatment indicator from the information provided by Helbig & Nikolai (2015) that shows whether or not children who dropped out of the high track received a low-track school degree.¹¹

The NEPS data provide detailed information on the schooling histories of a representative sample of adults living in Germany. The data have two specific features that enable us to tackle our research question in a methodologically clear-cut manner. First, the data include information on the nature of dropping out: that is, we can differentiate between whether a student left a school for good, whether the student switched to another school (i.e., another school track), and at what point in time the student effectively ended her schooling career.¹² Second, the NEPS data report the state in which the individual attended school. This is crucial for an analysis of reforms in a context where migration across states during an individual's lifespan is possible.

From the NEPS adult cohort we use a subsample of 3,878 adults who attended secondary school in a West German state (including Berlin) and were born between 1944 and 1983. In 2010, NEPS asked all participants about their school careers, documenting school types and locations of schools. We use these spell data to construct our main outcome variables, which are intended to capture important characteristics of school careers. Table 2 shows descriptive statistics for these outcome variables. The first variable *years of schooling* states the number

¹⁰See Blossfeld et al. (2011) for more detail.

¹¹The indicator does not distinguish between basic and middle school degrees as the type of degree varies greatly by state thus it would have been too complex to appropriately document this difference (Helbig & Nikolai 2015).

¹²This is in contrast to other studies that rely on school data reporting how many students left a school without a degree but have no information about the subsequent educational path of the reported dropout.

of years a person reported being in educational spells. The variable *longer in education than standard* is an indicator equal to 1 when a person stayed in the education system longer than would have been expected given her final degree. It is the difference between *years of schooling* and calculated standard years of schooling from the last CASMIN.¹³ The variable *no degree in last school episode* is an indicator equal to 1 when a person leaves the school system at a maximum of age 19 without reporting having obtained a degree in the last episode. When a person reports having obtained a basic school degree in the last school episode, the indicator *basic school degree in last school episode* is equal to 1. *Downgrading* is an indicator equal to 1 when a person switches from the high track to a lower-ranked school, that is, basic or middle school.¹⁴ Similarly, *upgrading* indicates whether a person switched from a lower-ranked school type to the high track. The *number of school episodes* counts how many schools a person attended and *age at leaving school system* is the age at which a person switched from school episodes to vocational training (including university education), employment, military service, or other types of spells.

The variables for school degrees are indicators as well, capturing reported school degrees at two different points in an individual's life. The first point is at the end of the last school episode, that is, between ages 11 and 19. The second point is in 2010, which should capture highest achieved school degree. The indicators differ as people can obtain school degrees later in life.

The occupational outcomes capture whether an individual holds a vocational or university degree in 2010 and the logarithm of the net household income in 2010. The vocational degree indicator is constructed from the CASMIN.

From the background questionnaire we construct control variables, including: sex, birth cohort, migration background, highest level of parental education, and number of siblings. We

¹³The CASMIN (Comparative Analysis of Social Mobility in Industrial Nations) has collected statistics about education since the 1980s.

¹⁴We cannot distinguish between these school types in the spell data.

construct three-year bins as birth cohort dummies - each including three successive birth cohort years.¹⁵ Table 2 contains summary statistics for the outcome and control variables.

4 Identification Strategy

To identify the causal effect of the dropout reform, we exploit the different timing of reform implementation between 1965 and 1999 in the 11 former West German states and employ a DiD approach. Our DiD approach relies on a comparison of students in the highest track preand post-reform within and between states. For the baseline specification, we restrict the sample to include only students who had a high-track episode during their school career. One issue that arises in evaluating reforms in general is selection into treatment. As soon as people know that the high track can grant a lower-track school degree, selection into the high track could change. To avoid selection effects, our treatment is not only the interaction of age and attending at least the high track once in life; we define our treatment group as high-track students who are 11-19 years old in the reform year. This restriction puts into the control group 93 students who chose to attend the high track after the reform at an age older than 11. Figure 3 shows our treatment and control groups for the 11 West German states. Black represents the cohorts at age 11 that are treated. White represents the pre-reform control group. Gray is the treatment group we exclude due to possible selection effects. In our main specification we estimate:

$$Y_{i,s,t} = \beta_0 + \beta_1 post \ reform_s + X_i\beta_5 + \mu_s + \mu_t + \varepsilon_{i,s,t}.$$
(1)

 $Y_{i,s,t}$ are the various outcome measures for school- and occupational-career characteristics of student *i* in state *s* of birth cohort *t*. The coefficient of interest is β_1 , showing the effect of the reform for students who attend the academic track in the reform year. The matrix X_i contains all control variables. State and birth cohort fixed effects are captured by μ_s and μ_t , respectively. In our main specification, standard errors are clustered at the state level as the treatment varies

¹⁵We do so because the number of observations per year of birth is small. Including dummies for each year of birth does not change the results.

at that level.¹⁶

The coefficient β_1 represents the causal effect of the reform if the assumptions of a DiD framework hold. Note that, in our setting, the common trend assumption has to hold, which implies that – in absence of the treatment – treatment and control groups would both lie on the same longer-run trend with respect to the outcome variables. The DiD literature often argues that this assumption holds with the same pre-trends before the reform. The NEPS data do not allow a thorough evaluation of pre-reform trends as the number of observations per birth cohort and state are small and decrease with age of the survey participants. Shares of people with a specific school degree, therefore, fluctuate a great deal by birth cohort and state. In Section 7 we provide several robustness checks of our identification strategy. We perform placebo tests, include linear state trends, and leave one state out at the time. A second crucial assumption of our approach is that the treatment effect we capture with β_1 should not represent any development other than the reform. To avoid capturing other developments within states that occurred at the same time as the reform under study, we investigated whether other education reforms affecting secondary schooling were introduced simultaneously and found no evidence that there were.

5 Main Results

In this section we describe and discuss the results from estimating Equation 1. First, we consider the reform's direct effects on the school careers of affected students. Second, we estimate effects on the school degrees they obtain. Third, we estimate whether there are any longer-run effects on occupational outcomes. The effects reported in this section are average treatment effects for the group of high-track students. Finally, we estimate the reform's effect on selection into the highest track.

¹⁶As we have only 11 German states, the number of clusters is small. We apply wild-cluster-bootstrap methods to account for this.

5.1 School Careers and Degrees

We expect to find that the reform resulted in less switching between school tracks by high-track students because, after the reform, they can obtain a lower-track school degree while attending the high-track school. This feature of the reform should reduce down- and upgrading. As to time spent in school, number of school episodes, and age at leaving the education system, the reform effect depends on the relative role of the following two incentives: (1) the incentive to leave school and enter the labor market before obtaining a high-track school degree and (2) the incentive to stay in the high-track school and try to finish 13 years of schooling with less risk of leaving school without any degree.

Table 3 shows that, overall, the reform affected the school careers of students in the highest track as expected, in that their school careers became more stable and linear. Students spend almost one year more in school (Column 1), which is about 70% of a standard deviation, meaning that they are about half a year older when leaving the school system (Column 3). It is not likely that this is due to students repeating grades more often after the reform as excess time in the highest track does not increase significantly (Column 2). These findings together indicate that students drop out later or finish school more often due to the reform. This is in line with a stronger incentive for trying to finish high-track school at lower risk. Additionally, school careers stabilize: students reduce their school episodes (Column 4) by half an episode, which is 60% of a standard deviation. This means that they switch less often between different schools. This is also true for downgrading, that is, switching to a lower-track school (Column 5). Downgrading is reduced by almost 90% of a standard deviation. Upgrading is slightly reduced as well, albeit not significantly. As students downgrade less often they logically cannot upgrade to the highest track as often as before. These results are robust to wild-cluster bootstrapping of the standard errors, as recommended by Cameron & Miller (2011). The third row of the table shows the p-value for testing whether the reform coefficient is zero, obtained from wild-cluster bootstrapping the standard errors. The significance levels hardly change when employing wildcluster bootstrapping methods.

The treatment group contains students aged 11 to 19. Results thus might be driven by a specific age group of students as it is possible, for example, that the incentive to leave school and enter

the labor market earlier is stronger for the older students than for the younger students. If this is the case, we would expect that the effects we find are driven by the younger students and might go into another direction for the older ones. Table 14 in the Appendix demonstrates that this is not the case. The effects always go in the same direction for all three age groups: ages 11-14, ages 15-17, and ages 18 and older.¹⁷ The significance levels are also almost the same for the different age groups. This suggests that the reform unlikely incentivized students to enter the labor market immediately.

One of the reform's goals was to provide students who otherwise would not have obtained a school degree with a school degree. Therefore, we expect that the share of people without any school degree decreased under the reform while the share of people with school degrees should increase (particularly for low- or middle-track school degrees). Table 4 sets out the reform's estimated effects on the likelihood of holding various school degrees. Columns 1 to 4 use information from the retrospective school spells about school degree obtained in the last reported school episode. Columns 5 to 8 use information on the highest school degree obtained by 2010. Column 1 shows that the likelihood to end the last reported school episode without any school degree is slightly, yet not significantly, reduced. This suggests that even before the reform, high-track dropouts rarely completed their school career without obtaining any degree. It is more likely and in line with anecdotal evidence that high-track dropouts obtained a lowertrack school degree at a lower-track school and finished their school career with such a degree. Indeed, we find that after the reform, students are less likely to finish their school careers with either a basic school degree (Column 2) or a middle school degree (Column 3). They are, in fact, more likely to finish their school career with a high-track school degree (Column 4). The same pattern holds for the highest school degree obtained by 2010 (see Columns 4 to 8). People who attended the highest track in the reform year hold significantly fewer middle school degrees but the share of people with a high-track degree increased by 60% of a standard deviation compared to people attending the high track before the reform. Again, the significance levels hold when applying wild-cluster bootstrapping methods (third row). In sum, Table 4 provides evidence

¹⁷The exception is upgrading, which, of course, is very unlikely for the older students as they would have to downgrade first and then upgrade.

that people are incentivized to stay in the high track and try to obtain a high-track school degree with a reduced risk of ending- up without any school degree.

The reform thus seems to motivate students to stay in the high track even if they are at risk of not completing it. Staying in the high track is surprisingly beneficial for students. As the estimates show, many of them actually obtain the high-track school degree and thus qualify for university and, consequently, a wider choice of career paths. Therefore, in the next section we investigate whether there the reform had longer-run effects on occupational outcomes.

5.2 Occupational Outcomes

More years of schooling and higher degrees should have a positive effect on students' labor market careers. Table 5 shows the reform's effect on occupational outcomes. Because the reform pushed students away from obtaining lower school degrees, the students are also less likely to hold a vocational school degree (Column 1) but more likely to obtain a university degree (Column 2). This push into higher education might be responsible for the higher net household income we observe in 2010 (Column 3). The first observation period in 2010 is not the optimal point in time to measure monetary returns to the reform as the control group in our sample is older and therefore more likely to be retired than the treatment group. Therefore, in the estimation equation of Column 3 we control for age and age squared in addition to our usual controls.

5.3 Selection into the Academic Track

As the reform reduced the risk of dropping- out of the high track without any school degree, we expect that (lower-ability) students should select into the high track. Lavecchia et al. (2016) and Woelfel & Heineck (2012) show that risk aversion plays an important role in schooling decisions. Relaxing our sample restriction from above to include students who attended the high track in the reform year, we analyze whether the share of students attending the high track at different points of their school career increased with the reform. Table 6 shows that after the reform, a higher share of students attended the high track – independent of measuring attending the high track as a dummy for having at least one high-track school episode, as attending the

high track as the first secondary school, or as attending the high track in eighth grade. However, the positive effect is only slightly significant for the likelihood of having any high-track school episode.¹⁸ The effect size is small; 5 percentage points, which is about a 10th of a standard deviation. One explanation for the small selection effects into the high track after the reform might be that at the time of the reform, parents were rarely allowed to choose which secondary school track their children would attend. Only in Bremen and Lower Saxony, and only since the late 1970s, did parents have more freedom of choice in this matter.

6 Reform Effect on Equality of Opportunity

The reform was intended to provide high-track dropouts with better labor market opportunities. Estimating average treatment effects, therefore, might attenuate the effect the reform has on the target population: those students at risk of dropout. Ex-post it is hard to define the group at risk because not all members of that group actually drop out. Indicators such as grades or repeating a grade are not documented in our data. In the following, we investigate the extent to which students with parents without a high-track degree profited from the reform differently than students with parents with a high-track degree. Dropping- out of the high track might be especially likely for those children who do not have strong parental support to pursue a high-track school career. Therefore, the group of students with parents without a high-track degree might belong to the reform's target population. However, there are only a few students with parents who hold a high-track degree.

Comparing the results in Panel A of Table 7 with those in Panel B shows that all coefficients go in the same direction for both groups. Also, the magnitude is mostly comparable except for excess time in education (Column 2), which is negative and insignificant for students of parents with a high-track school degree. Overall, effects are larger and therefore more significant for students with parents who do not hold a high-track degree.

¹⁸When we apply robust standard errors instead of clustered ones, the effects are significant for all three measures.

To detect differences in final school degrees, Panel A of Table 8 needs to be compared to Panel B. Again, the effects go in the same direction for both groups. Yet, the other estimates – particularly the reform's estimated effect on achievement of a high-track school degree (Columns 4 and 8) – is twice as large for students with parents without a high-track school degree. These results suggest that the reform was especially beneficial for students with parents who do not hold a high-track school degree.

7 Robustness Checks and Mechanisms

In this section we address some identification issues of our DiD approach and relax our sample restriction. First, we perform placebo tests, second, we include linear state-time trends, and, third, we exclude one state at a time. Fourth, we do not restrict the sample to high-track school students and include everybody who is 19 or younger in the year of the reform in the treatment group. Fifth, we investigate the reform effect for the middle- and basic-track students only, and look at the heterogeneity of the effect size for different age groups of high-track students.

7.1 Placebo Test

One major issue with our identification strategy is that the effect might not be driven by the specific reform but by other developments we do not observe. As we only compare means before and after the reform, it is possible that even developments that do not exactly coincide in timing with the reform under study are driving our results. To check whether this is the case, we estimated placebo reform effects for 10, 5, and 2 years prior to the reform in each state. Table 9 shows placebo reform effects for 10 years prior to the reform in Panel A, for 5 years prior to the reform in Panel B, and for 2 years prior to the reform in Panel C. The pattern we observe for the true reform year cannot be observed in any other year. When the coefficients are significant, the effect goes in the other direction. Most effects, however, are very close to zero. Although our sample is reduced by half - as earlier birth cohorts are less densely sampled - standard errors stay fairly comparable to the standard errors of our main results. These findings confirm that we are not capturing -any other unobserved developments with the reform effect and provide

evidence that we identify the true reform effect.

7.2 Linear State Trends

In DiD approaches it is possible that the reform effect is capturing -state-specific time trends and, if so, the effect cannot be interpreted as a causal one. To ensure that state-specific time trends are not driving our results we include linear state trends in Equation 1. The results are shown in Table 10. The table shows that our main findings of increased stability of school careers and a push into more high-track degrees are robust to the inclusion of linear state trends.¹⁹

7.3 Leave One Out

German education policy, in general, varies by state. It is thus possible that our reform effects are driven by specific states. To ensure that no single state is driving our results, we estimate all regressions in a restricted sample leaving one state out at a time. We report results only for the largest states, which contribute most observations to our analysis: Northrhine-Westphalia and Bavaria. Additionally, we show results for leaving out Berlin, a part of which was in East Germany at the time of the reform. Panel A of Table 11 shows the results for years of schooling and the number of school episodes. Panel B shows the results for downgrading and holding a high-track degree. The magnitudes of the coefficients hardly change and the effects stay significant. This holds as well for leaving out any other state (not shown here).

7.4 Effect on the Whole Population

To this point, we have identified reform effects for only a very small subgroup of the whole population of students within a state, namely, students in the high track. There are two arguments in favor of estimating an overall effect. First, the reform also targeted students in middle-track schools who finished the ninth grade and granted them a basic school degree. Second, there might be spillover effects on students from other school tracks. Resource constraints could pre-

¹⁹We can also include non-linear state-decade- fixed effects with the first decade of people born before 1950, the second people born between 1950 and 1959, the third people born between 1960 and 1969, the fourth people born between 1970 and 1979, and the fifth people born after 1979. These decade dummies are interacted with the state dummies. Our results do not change if we include the state-decade dummies.

vent high-track schools from becoming over-enrolled, whereas middle-track schools may try to attract students from the low-track schools to fill their seats. On the local labor markets it is possible that high-track school dropouts with a school degree crowd- out students from the middle-track schools. Table 12 shows that for the whole population the share of people without any school degree decreased significantly with the reform; all other effects, however, are close to zero and insignificant. This suggests that effects might go in different directions for different subgroups of students.

7.5 Heterogeneity and Subsamples

Finally, we investigate the reform's effect on the subgroup of students in the middle- and basictrack schools, as well as which age groups drive our effect for high-track students. Table 13 shows that the subsample of students who never attended the high track is larger than the hightrack sample by about 2,500 individuals. For these students, the reform effects are very close to zero and only significant for downgrading. First, this is evidence that there are no other reforms or unobserved developments coinciding with our reform that affect all students in general. Second, high-track students are the relevant treatment group and our reform effect can be interpreted as a treatment effect on the treated.

The treatment group consists of students between 11 and 19 years of age. Students in 10th grade or higher could leave the high track with a basic or middle school degree immediately after the reform, whereas students below 10th grade still had to successfully finish at least ninth grade. This difference in the treatment might lead to differences in the reform effect. Older students might be incentivized to leave school early while younger students might be encouraged to stay. In Table 14 we distinguish between three age groups: students aged 11 to 14, aged 15 to 17, and aged 18 or older. The results show that the effects go in the same direction for all age groups.²⁰ We interpret these findings as meaning that the reform incentivized staying in school longer in general.

²⁰The exception is *upgrading*, which shows a positive coefficient for the reform dummy. However, students aged 18 or older would not down-grade and then upgrade back to the high track anymore. It is plausible that the effect is zero for this group.

8 Conclusion

In the context of the German school system, we investigate the effects of a school reform that rewarded high-track school dropouts with a low-track school degree if they at least finished ninth grade. For identification, we exploit the fact that the reform was introduced in 11 West German states at different points during 1965 to 1982. Detailed retrospective spell data on school and labor market careers from the NEPS allow us to identify exactly who attended the high track, regardless of receiving a degree, and in which federal state. Both are features many other data sets lack. Our results show that the reform was successful, at least in the sense that students are less likely to switch between schools and tracks.

Surprisingly, however, we also find that the reform led to an increase in the number of high-track students who successfully finish high school and enter university. We furthermore find a positive effect on the students' income in later adult life. This is in line with theoretical considerations on risk perception that changes students' motivation: the reform seems to have reduced the perceived risk of trying longer in the high-track school and thus led to an increase in the high-track completion rate. The reform thus can be interpreted as a low-cost intervention that had a beneficial effect on students.

It could be argued that this study's very specific context limits the international policy conclusions that can be made. The dropout reform took place in a system with tracking and grade repetition, and where having a degree hold strong implications for labor market entrance. Thus, this reform might not work in another country's school system. However, our results reveal one important insight that is of importance to education policies in general: along the school career, rewarding each milestone achieved can motivate students to try longer and harder for success with the next milestone. A prominent example of this concept is the European Bologna reform that introduced the bachelor degree as a reward for university graduation. The results of our evaluation are evidence that rewarding milestones achieved can indeed change students' behavior and improve their student and life outcomes.

References

- Acemoglu, D. & Angrist, J. (2001), How Large are Human-Capital Externalities? Evidence from Compulsory-Schooling Laws, *in* 'NBER Macroeconomics Annual 2000, Volume 15', NBER Chapters, National Bureau of Economic Research, Inc, pp. 9–74.
 URL: https://ideas.repec.org/h/nbr/nberch/11054.html
- Anderson, D. M. (2014), 'In School and Out of Trouble? The Minimum Dropout Age and Juvenile Crime', *The Review of Economics and Statistics* 96(2), 318–331.
 URL: https://ideas.repec.org/a/tpr/restat/v96y2014i2p318-331.html
- Andrietti, V. (2015), The causal effects of increased learning intensity on student achievement: Evidence from a natural experiment, EconStor Preprints 120874, ZBW - German National Library of Economics.

URL: https://ideas.repec.org/p/zbw/esprep/120874.html

- Biewen, M. & Tapalaga, M. (2017), 'Life-cycle educational choices in a system with early tracking and second chance options', *Economics of Education Review* 56(Supplement C), 80–94.
- Blossfeld, H.-P., Rossbach, H.-G. & von Maurice, J. (2011), 'Education as a Lifelong Process: The German National Educational Panel Study (NEPS).', *Zeitschrift für Erziehungswissenschaft* 14.
- Buchholz, S. & Schier, A. (2015), 'New game, new chance? social inequalities and upgrading secondary school qualifications in west germany', *European Sociological Review* 31(5), 603–615.
- Cameron, A. C. & Miller, D. L. (2011), *Handbook of Empirical Economics and Finance*, CRC Press, chapter Robust Inference with Clustered Data, pp. 1–28.
- Dahmann, S. & Anger, S. (2015), The impact of education on personality : evidence from a German high school reform, IAB Discussion Paper 201429, Institut fuer Arbeitsmarktund Berufsforschung (IAB), Nuernberg [Institute for Employment Research, Nuremberg,

Germany].

129.

URL: https://ideas.repec.org/p/iab/iabdpa/201429.html

- Dearden, L., Emmerson, C. & Meghir, C. (2009), 'Conditional Cash Transfers and School Dropout Rates', *Journal of Human Resources* 44(4).
 URL: https://ideas.repec.org/a/uwp/jhriss/v44y2009i4p827-857.html
- DesJardins, S. L., Ahlburg, D. A. & McCall, B. P. (2006), 'The effects of interrupted enrollment on graduation from college: Racial, income, and ability differences', *Economics of Education Review* 25(6), 575–590.
- Dustmann, C., Puhani, P. A. & Schoenberg, U. (2017), 'The Longâterm Effects of Early Track Choice', *Economic Journal* 127(603), 1348–1380.
 URL: https://ideas.repec.org/a/wly/econjl/v127y2017i603p1348-1380.html
- Eyles, A., Hupkau, C. & Machin, S. (2016), 'School reforms and pupil performance', *Labour Economics* 41(C), 9–19.
 URL: https://ideas.repec.org/a/eee/labeco/v41y2016icp9-19.html
- Golsteyn, B. H., Groenqvist, H. & Lindahl, L. (2014), 'Adolescent Time Preferences Predict Lifetime Outcomes', *Economic Journal* 124(580), 739–761.
 URL: https://ideas.repec.org/a/wly/econjl/v124y2014i580pf739-f761.html
- Hanushek, E. A. & Woessmann, L. (2006), 'Does Educational Tracking Affect Performance and Inequality? Differences- in-Differences Evidence Across Countries', *Economic Journal* 116(510), 63–76.
 URL: https://ideas.repec.org/a/ecj/econjl/v116y2006i510pc63-c76.html
- Havnes, T. & Mogstad, M. (2011), 'No Child Left Behind: Subsidized Child Care and Children's Long-Run Outcomes', *American Economic Journal: Economic Policy* **3**(2), 97–

URL: https://ideas.repec.org/a/aea/aejpol/v3y2011i2p97-129.html

- Heckman, J. J., Humphries, J. E., LaFontaine, P. A. & Rodriguez, P. L. (2012), 'Taking the easy way out: How the ged testing program induces students to drop out', *Journal of Labor Economics* **30**(3).
- Helbig, M. & Nikolai, R. (2015), Die Unvergleichbaren: Der Wandel der Schulsysteme in den deutschen Bundesländern seit 1949., Klinkhardt.
- Huebener, M. & Marcus, J. (2017), 'Compressing instruction time into fewer years of schooling and the impact on student performance', *Economics of Education Review* 58(C), 1–14. URL: https://ideas.repec.org/a/eee/ecoedu/v58y2017icp1-14.html
- Hungerford, T. & Solon, G. (1987), 'Sheepskin Effects in the Returns to Education', *The Review* of Economics and Statistics **69**(1), 175–177.
- Jaeger, D. A. & Page, M. E. (1996), 'Degrees Matter: New Evidence on Sheepskin Effects in the Returns to Education', *The Review of Economics and Statistics* 78(4), 733–740.
- Kamhoefer, D. A. & Schmitz, H. (2016), 'Reanalyzing Zero Returns to Education in Germany', Journal of Applied Econometrics 31(5), 912–919.
 URL: https://ideas.repec.org/a/wly/japmet/v31y2016i5p912-919.html
- Lavecchia, A., Liu, H. & Oreopoulos, P. (2016), Chapter 1 behavioral economics of education: Progress and possibilities, Vol. 5 of *Handbook of the Economics of Education*, Elsevier, pp. 1 – 74.
- Layard, R. & Psacharopoulos, G. (1974), 'The Screening Hypothesis and the Returns to Education', *Journal of Political Economy* **82**(5), 985–998.
- Levitt, S. D., List, J. A., Neckermann, S. & Sadoff, S. (2016), 'The Behavioralist Goes to School: Leveraging Behavioral Economics to Improve Educational Performance', *American Economic Journal: Economic Policy* 8(4), 183–219.
 URL: https://ideas.repec.org/a/aea/aejpol/v8y2016i4p183-219.html

- Meghir, C. & Palme, M. (2005), 'Educational Reform, Ability, and Family Background', *American Economic Review* 95(1), 414–424.
 URL: https://ideas.repec.org/a/aea/aecrev/v95y2005i1p414-424.html
- Oreopoulos, P. (2007), 'Do dropouts drop out too soon? wealth, health and happiness from compulsory schooling', *Journal of Public Economics* **91**, 2213–2229.
- Peisert, H. & Dahrendorf, R. (1967), Der vorzeitige Abgang vom Gymnasium. Studien und Materialien zum Schulerfolg an den Gymnasien in Baden-Württemberg 1953-1963, Bildung in neuer Sicht.
- Pekkarinen, T., Uusitalo, R. & Kerr, S. (2009), 'School tracking and intergenerational income mobility: Evidence from the Finnish comprehensive school reform', *Journal of Public Economics* 93(7-8), 965–973.

URL: https://ideas.repec.org/a/eee/pubeco/v93y2009i7-8p965-973.html

- Piopiunik, M. (2014), 'The effects of early tracking on student performance: Evidence from a school reform in Bavaria', *Economics of Education Review* 42(C), 12–33.
 URL: https://ideas.repec.org/a/eee/ecoedu/v42y2014icp12-33.html
- Pischke, J.-S. (2007), 'The Impact of Length of the School Year on Student Performance and Earnings: Evidence From the German Short School Years', *Economic Journal* 117(523), 1216–1242.

URL: https://ideas.repec.org/a/ecj/econjl/v117y2007i523p1216-1242.html

- Pischke, J.-S. & von Wachter, T. (2008), 'Zero Returns to Compulsory Schooling in Germany:
 Evidence and Interpretation', *The Review of Economics and Statistics* 90(3), 592–598.
 URL: https://ideas.repec.org/a/tpr/restat/v90y2008i3p592-598.html
- Riphahn, R. T. (2012), 'Effect of Secondary School Fees on Educational Attainment', *Scandinavian Journal of Economics* 114(1), 148–176.
 URL: https://ideas.repec.org/a/bla/scandj/v114y2012i1p148-176.html

- Siedler, T. (2010), 'Schooling and Citizenship in a Young Democracy: Evidence from Postwar Germany', Scandinavian Journal of Economics 112(2), 315–338.
 URL: https://ideas.repec.org/a/bla/scandj/v112y2010i2p315-338.html
- Thiel, H., Thomsen, S. L. & Buettner, B. (2014), 'Variation of learning intensity in late adolescence and the effect on personality traits', *Journal of the Royal Statistical Society Series* A 177(4), 861–892.

URL: https://ideas.repec.org/a/bla/jorssa/v177y2014i4p861-892.html

Woelfel, O. & Heineck, G. (2012), 'Parental risk attitudes and children's secondary school track choice', *Economics of Education Review* **31**(5), 727–743.

Tables and Figures



Figure 1: The Structure of the German School System



Figure 2: Dropouts from the high track in Germany

Notes: The blue dots show for birth cohorts from 1950 to 1986 the share of people who attended the high track in Germany at age 13/14 (8th grade). The orange dots shows for the same birth cohorts the share of people who dropped out from the high track among the people who attended the high track in 8th grade. The respective lines are the linear fit of the raw data points. Data source: NEPS SC6:7.0.0. Own calculations.



Figure 3: Restricted sample: control and treatment group sample

Notes: The figure shows the restricted sample we use for our main analysis. The cohorts which turn 11 in the specific year are pre-reform cohorts if the cell is white and belong treatment group not prone to selection if the cell is black. Grey cells are the birth cohorts which belong to the treatment group prone to selection and are excluded from our analysis. Source: (Helbig & Nikolai 2015).

state	reform
Baden Wuerttemberg	1965
Hamburg	1966
Hessen	1967
Schleswig Holstein	1970
Rheinland Pfalz	1972
Niedersachsen	1976
Saarland	1976
Berlin	1977
Nordrhein Westfalen	1978
Bayern	1982
Brandenburg	1991
Mecklenburg Vorpommern	1994
Bremen	1996
Sachsen	1999
Sachsen Anhalt	2002
Thueringen	2003

Table 1: Timing of the dropout reforms

Notes: The table shows the timing of the dropout reforms in the German federal states. Source: (Helbig & Nikolai 2015).

Table 2: Summary statistics

	Mean	SD	Min	Max
School career				
Vears of schooling	12 30	1.00	8	13
Exacts time in education	0.21	0.46	0	15
	0.51	0.40	14	1
Age at leaving school system	18.55	0.90	14	19
School episodes	2.56	0.85	1	8
Downgrading from high track	0.13	0.33	0	1
Upgrading into high track	0.12	0.32	0	1
School degree				
No degree in last school year	0.05	0.21	0	1
Basic school degree in last school year	0.02	0.14	Ő	1
Middle ashaal dagaa in last school year	0.02	0.14	0	1
Middle school degree in fast school year	0.19	0.39	0	1
High track school degree in last school year	0.69	0.46	0	1
No degree in 2010	0.00	0.03	0	1
Basic school degree in 2010	0.01	0.11	0	1
Middle school degree in 2010	0.15	0.36	0	1
High track school degree in 2010	0.83	0.37	0	1
	0.85	0.37	0	1
Occupational Outcomes				
University degree	0.54	0.50	0	1
Holding a vocational degree	0.40	0.49	0	1
Net household income (in logs)	9.264	0.674	0	11.717
Personal characteristics				
Formala	0.40	0.50	0	1
remaie	0.49	0.50	10.44	1002
Year of birth	1958.22	6.23	1944	1982
Migration background	0.08	0.26	0	1
Highest parental education				
No degree	0.00	0.05	0	1
Basic school degree	0.42	0.49	0	1
Middle school degree	0.12	0.12	Ő	1
	0.20	0.70	0	1
High track school degree	0.11	0.32	0	1
University degree	0.27	0.44	0	I
Number of siblings				
No siblings	0.16	0.37	0	1
One sibling	0.38	0.49	0	1
Two or more siblings	0.46	0.50	0	1
Location of secondary school				
Schleswig-Holstein	0.03	0.16	0	1
Homburg	0.03	0.10	0	1
namourg	0.01	0.07	0	1
Lower Saxony	0.12	0.33	0	1
Bremen	0.04	0.19	0	1
Northrhine-Westphalia	0.36	0.48	0	1
Hesse	0.03	0.17	0	1
Rhineland-Palatinate	0.05	0.21	0	1
Roden Wuerttemberg	0.05	0.21	0	1
Dauen-wueluenibeig	0.03	0.22	0	1
Bavaria	0.25	0.43	0	1
Saarland	0.03	0.17	0	1
Berlin	0.04	0.19	0	1
Observations	1462			

Notes: The table shows summary statistics for the outcome and control variables constructed from the NEPS data.

	(1) Years of schooling	(2) Excess time in education	(3) Age at leaving school system	(4) School episodes	(5) Downgrading	(6) Upgrading
post reform	0.725*** (0.121)	0.055 (0.050)	0.605*** (0.076)	-0.546*** (0.111)	-0.292*** (0.039)	-0.032 (0.035)
p-value wild-cluster bootstrap	[0000]	[0.476]	[000.0]	[0.004]	[0.004]	[0.48]
state FE	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes
observations	1462	1461	1460	1462	1460	1460
R^2	0.105	0.065	0.087	0.094	0.098	0.122
mean outcome	12.296	0.313	18.547	2.563	0.128	0.120
SD outcome	1.088	0.464	0.895	0.851	0.334	0.325

Table 3: Reform effect on school career

downgrading, and upgrading described in detail in section 3. We report the coefficient *post reform* which is the reform effect for students age 11-19 attending the high track in the year of the reform. Standard errors are clustered on state level. The third row reports p-values of wild-cluster bootstrapping the standard errors. All specifications include ber of school episodes, state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Asterisks show the significance level: * = p < 0.1, ** = p < 0.05, *** = p < 0.01. Notes: The table shows the res

	q	egree in las	t school epis	sode		final schoo	ol degree 20	10
	(1) none	(2) basic	(3) middle	(4) high track	(5) none	(6) basic	(7) middle	(8) high track
post reform	-0.026	-0.037**	-0.256***	0.359***	0.002	-0.005	-0.235***	0.238***
p-value wild-cluster bootstrap	[0.456]	[0.184]	[0.004]	[000.0]	[0.372]	[0.688]	[0.004]	[0000]
state FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
observations	1460	1460	1460	1460	1462	1462	1462	1462
R^2	0.057	0.025	0.079	0.094	0.012	0.013	0.081	0.079
mean outcome	0.047	0.019	0.188	0.693	0.001	0.013	0.155	0.832
SD outcome	0.212	0.137	0.391	0.461	0.026	0.113	0.362	0.374

Table 4: Reform effect on school degrees

in 2010 (no degree, basic, middle, and high track school degree) described in detail in section 3. We report the coefficient *post reform* which is the reform effect for students age 11-19 attending the high track in the year of the reform. Standard errors are clustered on state level. The third row reports p-values of wild-cluster bootstrapping the standard Notes: The table shows the results of estimating equation 1 for degrees obtained in the last school year (no degree, basic, middle, and high track school degree) and degrees held errors. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Asterisks show the significance level:* = p<0.1, **=p<0.05, ***=p<0.01.

	(1)	(2)	(3)
	Holding vocational degree	university degree	In(household income)
post reform	-0.140***	0.137^{***}	0.119*
	(0.037)	(0.038)	(0.065)
state FE	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes
observations	1462	1462	1444
R^2	0.091	0.072	0.039
mean outcome	0.399	0.540	9.264
SD outcome	0.490	0.499	0.674

Table 5: Reform effect on occupational outcomes

held at age 20 and 35. The variables are described in detail in section 3. We report the coefficient *post reform* which is the reform effect for students age 11-19 attending the high track in the year of the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level: * = p<0.1, **=p<0.05, ***=p<0.01. Notes: The table shows the results of estimating equation 1 for the likelihood of holding a vocational, or a university degree in 2010 and the quality of the job in terms of income

	(1) Any high track episode	(2) First secondary school: high track	(3) Grade 8: high track
post reform	0.050* (0.027)	0.034 (0.025)	0.025 (0.022)
state FE	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes
observations	8621	8233	8621
R^2	0.180	0.175	0.177
mean outcome	0.361	0.362	0.432
SD outcome	0.480	0.481	0.495

Table 6: Reform effect: selection into high track

Notes: For the overall population we estimate if the share of students in the high track increased after the reform. We distinguish between having any high track episode (column 1), reporting the high track as first attended secondary school (column 2), and attending the high track in 8th grade. *Post reform* captures if students are 19 or younger at the year of the reform. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level: * = p<0.1, **=p<0.05, ***=p<0.01.

	(1)	(2)	(3)	(4)	(5)	(9)
	Years of	Excess time	Age at leaving	School	Downgrading	Upgrading
	schooling	in education	school system	episodes		
Panel A. Subsam	uple: parents	with high track	school degree			
post reform	0.278	-0.090	0.258^{*}	-0.387^{**}	-0.156^{**}	-0.033
4	(0.191)	(0.063)	(0.118)	(0.172)	(0.063)	(0.040)
observations	557	556	557	557	557	557
R^2	0.108	0.108	0.091	0.090	0.085	0.109
mean outcome	12.454	0.300	18.662	2.542	0.110	0.065
SD outcome	0.954	0.459	0.740	0.882	0.313	0.246
Panel B. Subsam	uple: parents	without high tr	ack school degree			
post reform	0.962^{***}	0.126^{**}	0.800^{***}	-0.641***	-0.359***	-0.035
	(0.148)	(0.047)	(0.112)	(0.112)	(0.061)	(0.046)
observations	905	905	903	905	903	903
R^2	0.119	0.062	0.098	0.116	0.142	0.134
mean outcome	12.199	0.320	18.476	2.576	0.140	0.154
SD outcome	1.153	0.467	0.973	0.832	0.347	0.361
state FE	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes
of our subaroup	nalweie reetr	icting the same	le to people with	narents who	hold an high trac	t school deared

Table 7: Heterogeneity of reform effect on school careers by parental school degree

Notes: The table shows the results of our subgroup analysis restricting the sample to people with parents who hold an high track school degree. We estimate equation 1 for this subgroup with years of schooling, excess time in school, age at leaving the school system, the number of school episodes, downgrading, and upgrading as outcome variables described in detail in section 3. We report the coefficient *post reform* which is the reform effect for students age 11-19 attending the high track in the year of the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level: * = p<0.1, **=p<0.05, ***=p<0.01.

		degree in la	ast school ye	ar		final scho	ol degree 20	10
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	none	basic	middle	high track	none	basic	middle	high track
Panel A. Subsan	ıple: paren	ts with high	track schoo	l degree				
post reform	0.022	-0.038	-0.152***	0.211^{**}	0.005	-0.002	-0.139***	0.136^{**}
I	(0.017)	(0.032)	(0.033)	(0.074)	(0.003)	(0.014)	(0.037)	(0.048)
observations	557	557	557	557	557	557	557	557
R^2	0.105	0.061	0.081	0.108	0.029	0.033	0.096	0.087
mean outcome	0.047	0.020	0.142	0.738	0.002	0.011	0.113	0.874
SD outcome	0.211	0.139	0.349	0.440	0.042	0.103	0.317	0.332
Panel B. Subsan	nple: paren	its without h	ugh track sci	iool degree				
post reform	-0.053	-0.031^{***}	-0.304***	0.429^{***}	0.000	-0.007	-0.282***	0.289^{***}
I	(0.036)	(0.00)	(0.055)	(0.050)	:	(0.012)	(0.033)	(0.031)
observations	903	903	903	903	905	905	905	905
R^2	0.061	0.021	0.088	0.105		0.025	0.084	0.088
mean outcome	0.048	0.019	0.217	0.666	0.000	0.014	0.180	0.806
SD outcome	0.213	0.136	0.412	0.472	0.000	0.119	0.384	0.396
state FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table 8: Heterogeneity of reform effect on school degrees by parental school degree

Notes: The table shows the results of our subgroup analysis restricting the sample to people with parents who hold an high track school degree. We estimate equation 1 for this subgroup with degree obtained in the last school year (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree, basic, middle, and high track school degree) and degrees held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree held in 2010 (no degree) and the school degree) and the school degree) and the schoo school degree) described in detail in section 3. We report the coefficient *post reform* which is the reform effect for students age 11-19 attending the high track in the year of the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level:* = p<0.1, **=p<0.05, ***=p<0.01. Table 9: Placebo test

)	
	(1) Years of schooling	(2) School episodes	(3) Downgrading	(4) none	(5) basic	(6) middle	(7) academic
nel A. Placebo reforn	1 10 years earlier						
cebo reform effect	-0.090*	-0.131^{*}	0.027^{**}	0.001	-0.008	0.010	-0.004
	(0.042)	(0.055)	(0.00)	(0.002)	(0.011)	(0.027)	(0.039)
servations	620	620	618	620	620	620	620
	0.072	0.051	0.033	0.021	0.023	0.048	0.048
nel B. Placebo reforn	1 5 years earlier						
cebo reform effect	-0.001	-0.146	-0.010	-0.006	-0.023***	0.008	0.020
	(0.105)	(0.082)	(0.040)	(0.004)	(0.003)	(0.023)	(0.019)
servations	620	620	618	620	620	620	620
	0.071	0.050	0.032	0.022	0.024	0.048	0.048
nel C. Placebo reforn	n 2 years earlier						
cebo reform effect	-0.141	0.175^{**}	-0.022	0.006	0.001	0.029	-0.037
	(0.074)	(0.043)	(0.027)	(0.004)	(0.014)	(0.035)	(0.021)
servations	620	620	618	620	620	620	620
	0.072	0.051	0.032	0.023	0.022	0.048	0.049
te FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
thcohort FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
an outcome	12.27	2.642	0.138	0.002	0.016	0.155	0.827
outcome	1.066	0.910	0.345	0.040	0.126	0.362	0.378

Notes: The table shows the results of estimating equation 1 for our main outcome variables described in detail in section 3. The main effect of interest is the coefficient of *trend 10 years before reform* which is the reform effect for students age 11-19 attending the high track 10, 5, and 2 years prior to the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level:* = p<0.1, **=p<0.05, ***=p<0.01.

	(1) Years of schooling	(2) No. of school episodes	(3) Downgrading	(4) no degree	(5) basic school degree	(6) middle school degree	(7) academic school degree
st reform	0.823*** (0.072)	-0.548*** (0.130)	-0.339*** (0.028)	0.001 (0.001)	-0.005 (0.012)	-0.276*** (0.019)	0.280^{***} (0.024)
lear state trends	Yes	Yes	Yes	Yes	Yes	Yes	Yes
ate FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
rthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
servations	1462	1462	1460	1462	1462	1462	1462
	0.138	0.100	0.113	0.016	0.017	0.103	0.102
ean outcome	12.296	2.563	0.128	0.001	0.013	0.155	0.832
) outcome	1.088	0.851	0.334	0.026	0.113	0.362	0.374

ends
te tre
r stai
linea
uding
inclu
heck:
ess cl
bustne
: Ro
10
Table

Notes: The table shows robustness checks including in our estimation equation 1 linear time-state trends. The outcome variables are years of schooling, school episodes, downgrading, holding no degree, a basic, middle, or high track school degree in 2010. The variables are described in detail in section 3. We report the coefficient *post reform* which is the reform effect for students age 11-19 attending the high track in the year of the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level:* = p<0.1, **=p<0.05, ***=p<0.01.

1 mmc 11.	Ye	cars of schooli	ng	(C)	school episode	S
	(1)	(2)	(3)	(4)	(5)	(9)
	no NRW	no Bavaria	no Berlin	no NRW	no Bavaria	no Berlin
post reform	0.672^{***}	0.770^{***}	0.746^{***}	-0.530^{***}	-0.683***	-0.554***
	(0.187)	(0.156)	(0.124)	(0.133)	(0.151)	(0.111)
state FE	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes
observations	941	1095	1406	941	1095	1406
R^2	0.088	0.121	0.111	0.088	0.129	0.093
mean outcome	12.243	12.291	12.303	2.589	2.536	2.568
SD outcome	1.130	1.077	1.092	0.881	0.810	0.857
Panel B.						
		Downgrading		high	track school d	legree
	(1)	(2)	(3)	(4)	(5)	(9)
	no NRW	no Bavaria	no Berlin	no NRW	no Bavaria	no Berlin
post reform	-0.287***	-0.287***	-0.291^{***}	0.238^{***}	0.287^{***}	0.237^{***}
	(0.061)	(0.061)	(0.041)	(0.044)	(0.046)	(0.032)
state FE	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes
observations	939	1094	1404	941	1095	1406
R^2	0.105	0.084	0.096	0.088	0.083	0.077
mean outcome	0.140	0.103	0.130	0.815	0.853	0.828
SD outcome	0.347	0.304	0.337	0.388	0.354	0.378

Table 11: Robustness check: leave one state out at a time

downgrading, and holding a high track school degree. The variables are described in detail in section 3. Columns 1 and 4 show results for leaving out Northrhine-Westphalia, columns 2 and 5 show results for leaving out Bavaria, and column 3 and 6 show results for leaving out Berlin. We report the coefficient post reform which is the reform effect for /ears of schooling, school episodes, students age 11-19 attending the high track in the year of the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6.7.0.0. Standard errors are clustered on state level. Asterisks show the significance level: * = p < 0.1, ** = p < 0.05, *** = p < 0.01. Notes: The table shows robustness ch

		school career		IJ	inal school	l degree 2(010
	(1) Years of schooling	(2) School episodes	(3) Downgrading	(4) none	(5) basic	(6) middle	(7) academic
post reform	-0.118 (0.146)	0.027 (0.047)	0.003 (0.008)	-0.001 (0.003)	0.011 (0.029)	0.005 (0.020)	-0.015 (0.022)
state FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
observations	10032	10032	9920	10032	10032	10032	10032
R^2	0.206	0.066	0.016	0.021	0.102	0.065	0.176
mean outcome	11.237	2.521	0.056	0.006	0.181	0.315	0.498
SD outcome	1.608	0.866	0.230	0.078	0.385	0.464	0.500

Table 12: Robustness check: effect on whole population

Notes: The table shows the reform effect on school careers and school degrees for the whole population (not only high track attendees). The coefficient of *post reform* is the treatment effect for students being younger than 19 in the year of the reform. All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level:* = p<0.1, **=p<0.05, ***=p<0.01.

		school career		f	inal school	l degree 20	010
	(1) Years of schooling	(2) School episodes	(3) Downgrading	(4) none	(5) basic	(6) middle	(7) academic
post reform	0.043 (0.108)	-0.005 (0.039)	-0.016** (0.007)	0.006 (0.004)	-0.062 (0.054)	0.030 (0.045)	0.025 (0.024)
state FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
birthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
observations	2564	2564	2538	2564	2564	2564	2564
R^2	0.084	0.061	0.092	0.030	0.089	0.067	0.089
mean outcome	10.258	2.511	0.016	0.006	0.304	0.473	0.217
SD outcome	1.309	0.754	0.126	0.076	0.460	0.499	0.412

Table 13: Reform effect on students in middle and basic school track

Notes: All specifications include state- and birth-cohort fixed effects and the set of control variables. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level: * = p < 0.1, ** = p < 0.05, *** = p < 0.01.

	(1)	(2)	(3)	(4)	(5)	(9)
	Years of schooling	Excess time in education	Age at leaving school system	School episodes	Downgrading	Upgrading
11-14 years post reform	0.560* (0.270)	0.054 (0.071)	0.387** (0.142)	-0.657*** (0.152)	-0.274*** (0.061)	-0.121*** (0.036)
5-17 years post reform	0.680*** (0.153)	0.037 (0.053)	0.599*** (0.092)	-0.636*** (0.124)	-0.293*** (0.049)	-0.076* (0.039)
18+ years post reform	0.796*** (0.102)	0.072 (0.052)	0.649^{***} (0.060)	-0.444*** (0.131)	-0.295*** (0.030)	0.024 (0.044)
state FE	Yes	Yes	Yes	Yes	Yes	Yes
pirthcohort FE	Yes	Yes	Yes	Yes	Yes	Yes
observations	1462	1461	1460	1462	1460	1460
զշ	0.108	0.066	0.093	0.097	0.098	0.131
nean outcome	12.296	0.313	18.547	2.563	0.128	0.120
SD outcome	1.088	0.464	0.895	0.851	0.334	0.325

Table 14: Reform effect on school careers by age group

Notes: In the table we estimate the reform effect on school careers for three different age groups of high track school attendees. The first age group is 11-14 years old in the year of the reform, the second 15-17, and the third 18 and older. Data stems from NEPS SC6:7.0.0. Standard errors are clustered on state level. Asterisks show the significance level:* = p<0.1, **=p<0.05, ***=p<0.01.